**Author’s response to reviews**

**Title:** Comparison of the CES-D and PHQ-9 depression scales in people with type 2 diabetes in Tehran, Iran

**Authors:**

Mohammad Ebrahim Khamseh (khamseh@iums.ac.ir)
Hamid Reza Baradaran (hrbaradaran@tums.ac.ir)
Anna Javanbakht (ajavanbakht225@gmail.com)
Maryam Mirghorbani (maryam.mirghorbani@yahoo.com)
Zahra Yadollahi (zyadollahi@yahoo.com)
Mojtaba Malek (malekmo@yahoo.com)

**Version:** 5  **Date:** 7 April 2011

**Author’s response to reviews:**

Dear Editor

I would like to inform you that all requested issues have been considered in this revision for publication.

With kind regards

Dr Mohammad E Khamseh